Author's response to reviews

Title: Parental beliefs & practice of spiritual methods for their sick children at a tertiary care hospital of Pakistan- A cross sectional questionnaire study.

Authors:

Ghulam Mustafa (falahchild@hotmail.com)
Nadir Bashir (dr.nadirbashir@gmail.com)
Asadi M Aslam (aslamasadi@bzu.edu.pk)

Version: 4 Date: 13 September 2015

Author's response to reviews: see over
Dear Editor,

Kindly find attached the manuscript entitled **Parental beliefs & practice of spiritual methods for their sick children at a tertiary care hospital of Pakistan- A cross sectional questionnaire study.**

Though many parents use the spiritual methods as a part of therapy for their children but there is no previous evaluation of the parental KAP about it. The parental practices for their children have never been focused before.

This is first of its kind.

Changes made as advised by the reviewers.

1. Page 6, Line 182; The ethical statement now includes retrospectively.
2. Major changes have been made throughout the text for grammar & language.

Hope that suffices now.

Regards,
Ghulam Mustafa